Author's response to reviews

Title: Does Alendronate reduce the risk of fracture in men? A meta-analysis incorporating prior knowledge of anti-fracture efficacy in women

Authors:

Anna M Sawka MD (sawkaam@yahoo.com)
Alexandra Papaioannou MD (jd.adachi@sympatico.ca)
Jonathan D Adachi MD (jd.adachi@sympatico.ca)
Amiram Gafni PhD (sawkaam@yahoo.com)
David A Hanley MD (jd.adachi@sympatico.ca)
Lehana Thabane PhD (sawkaam@yahoo.com)

Version: 5 Date: 28 June 2005

Author's response to reviews: see over
June 28, 2005

J.D. Adachi, MD
501-25 Charlton Ave. E.
Hamilton, Ontario, Canada, L8N 1Y2
Ph: (905) 529-1317 Fax: (905) 521-1297
E-mail: jd.adachi@sympatico.ca

To the Editor of BioMed Central Musculoskeletal Disorders,

We would again like to thank the editor and the statistical reviewer, Dr. Huson, for reviewing our revised manuscript, titled, “Does Alendronate Reduce the Risk of Fracture in Men? A meta-analysis incorporating prior knowledge of anti-Fracture efficacy in women” (MS 1079418594774445). We greatly appreciate our manuscript being provisionally accepted for publication. We have made the formatting changes that were requested in the newly revised manuscript. If you have any questions or concerns about the manuscript, please contact Dr. Adachi or A. Sawka

Sincerely,

A.M. Sawka, MD and J.D. Adachi, MD